Author's response to reviews

Title: Definition and characterization of localised meningitis epidemics in Burkina Faso: a longitudinal retrospective study

Authors:

Haoua Tall (htall@aamp.org)
Stéphane Hugonnet (hugonnets@who.int)
Philippe Donnen (philippe.donnen@ulb.ac.be)
Michèle Dramaix-Wilmet (miwilmet@ulb.ac.be)
Ludovic Kambou (kambouludo@hotmail.com)
Frank Drabo (drabofranck@yahoo.fr)
Judith E Mueller (judith.mueller@ehesp.fr)

Version: 5 Date: 30 December 2011

Author's response to reviews: see over
Dear editors,

We hereby submit the final version of the manuscript MS: 1669936829539695 – “Definition and characterization of localized meningitis epidemics in Burkina Faso: a longitudinal retrospective study”.

As you recommended, we have given consideration to all remarks by reviewer 1. A detailed response is included in this submission.

We also have replaced reference 12 by a more recent and more exhaustive publication on the same epidemic event. If the editors prefer, the older reference is valid, nevertheless.

Thank-you for agreeing to publish this manuscript.

Yours sincerely,

Judith Mueller, MD MPH PhD

EHESP

j Judith.mueller@ehesp.fr